Author’s response to reviews

Title: Induction of the interleukin 6/ signal transducer and activator of transcription pathway in the lungs of mice sub-chronically exposed to mainstream tobacco smoke.

Authors:

Sabina Halappanavar (sabina_halappanavar@hc-sc.gc.ca)  
Marsha Russell (marsha_russell@hc-sc.gc.ca)  
Martin R Stampfli (stampfli@mcmaster.ca)  
Andrew Williams (Andrew_williams@hc-sc.gc.ca)  
Carole L Yauk (carole_yauk@hc-sc.gc.ca)

Version: 7 Date: 30 July 2009

Author’s response to reviews:

Dear Editor,

At the stage of proof read, we found that the figures were wrongly numbered. For eg figure 6 was numbered figure 1. figure-1 was numbered figure-2 etc. Also the figure legends read figure-6 as figure-1. Our submitted maunscript shows the figures and figure leeneds in right order.

We have reloaded the files in correct order. I hope it is not too late to make this correction

thank you

sabina halappanavpr.